Author's response to reviews

Title: Randomized, controlled, parallel-group prospective study to investigate the clinical effectiveness of early insulin treatment in patients with latent autoimmune diabetes in adults.

Authors:

Sinead Brophy (s.brophy@swansea.ac.uk)
Helen Davies (h.davies@swansea.ac.uk)
Stephen Bain (s.c.bain@swansea.ac.uk)
Jeffrey W Stephens (j.w.stephens@swansea.ac.uk)
Wei-yee Cheung (w.y.cheung@swansea.ac.uk)
Kez Richards (kez.richards@swansea-tr.wales.nhs.uk)
Kathie Wareham (kathie.wareham@swansea-tr.wales.nhs.uk)
Charles Beaverstock (Charles.Beaverstock@bromor-tr.wales.nhs.uk)
Janet Lloyd (j.lloyd34@ntlworld.com)
Don Page (h.davies@swansea.ac.uk)
Meurig Williams (DrMeurig.Williams@carmarthen.wales.nhs.uk)
Ian Russell (ian.russell@bangor.ac.uk)
Rhys Williams (d.r.r.williams@swansea.ac.uk)

Version: 8 Date: 17 July 2008

Author's response to reviews:

Dear Editor

We have made the follow ammendments as requested:

1. Study protocol heading has been removed
2. Clinical trial number has been removed from title
3. Affiliations have been written in full.
4. Manuscript sections have been re arranged and a discussion has been added.
5. Sentence case has been used for section headings
6. Acknowledgments have been added
7. Figures have been cited
8. Figure titles has been relabeled as figure legends
9. Figure 1 has been cropped.

Thank you very much for all the work on this manuscript
With very good wishes and thanks

Sinead